Author’s response to reviews

Title: Maternal outcome of delivery among adolescent girls at the Yaounde Central Hospital

Authors:
Florent Ymele Fouelifack (yfouelifack@gmail.com)
Philip Njotang Nana (nananjotangphilip@yahoo.com)
Theodore Yangsi Tameh (theotameh@yahoo.co.uk)
Ngole Eta Mbong (mbongeta@yahoo.fr)
Hortence Jeanne Fouedjio (fouedjiojeanne@yahoo.fr)
Robinson Enow Mbu (rembu2000@yahoo.com)

Version: 3 Date: 16 August 2013

Author’s response to reviews:

To the Editor
BMC Pregnancy and Childbirth
Dear Sir/Madam

We hereby submit for your appreciation, the reviewed manuscript of the study we had earlier submitted with the following modifications:
1. The word ‘maternal’ in front of the word ‘outcome’ in the article title was deleted given the fact that we included aspects of fetal perinatal vitality mortality
2. The word ‘premature delivery’ was changed to ‘preterm delivery’, the correct term you proposed.
3. Elements of fetal vitality and mortality as well as maternal mortality were added as requested

We do appreciate your patience and extensions you gave to our team to improve upon the draft and add what you requested.

Kind regards,

Florent Ymele Fouelifack, MD
On behalf of the research team.